

Grounding Line Retreat of Denman Glacier, East Antarctica, Measured With COSMO-SkyMed Radar Interferometry Data

V. Brancato, E. Rignot, P. Milillo, M. Morlighem, J. Mouginot, L. An, B. Scheuchl, S. Jeong, P. Rizzoli, J. Bueso bello, et al.

▶ To cite this version:

V. Brancato, E. Rignot, P. Milillo, M. Morlighem, J. Mouginot, et al.. Grounding Line Retreat of Denman Glacier, East Antarctica, Measured With COSMO-SkyMed Radar Interferometry Data. Geophysical Research Letters, 2020, 47 (7), 10.1029/2019GL086291 . hal-03026293

HAL Id: hal-03026293 https://hal.science/hal-03026293v1

Submitted on 16 Apr 2021

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution - NoDerivatives 4.0 International License



Geophysical Research Letters

RESEARCH LETTER

10.1029/2019GL086291

Key Points:

- SAR observations of Denman Glacier grounding line, East Antarctica, reveal fast retreat of a glacier with a 1.5 m sea level rise equivalent
- Denman is retreating along a deep trough with a retrograde bed deepening up to 3.5 km below sea level, one of the deepest basin in Antarctica
- Warm water intrusion in the sub-ice shelf cavity and the retrograde bed topography makes Denman prone to marine instability in near future

Supporting Information:

Supporting Information S1

Correspondence to:

V. Brancato, virginia.brancato@jpl.nasa.gov

Citation:

Brancato, V., Rignot, E., Milillo, P., Morlighem, M., Mouginot, J., An, L., et al. (2020). Grounding line retreat of Denman Glacier, East Antarctica, measured with COSMO-SkyMed radar interferometry data. *Geophysical Research Letters*, 47, e2019GL086291. https://doi.org/10.1029/2019GL086291

Received 4 DEC 2019 Accepted 27 FEB 2020 Accepted article online 23 MAR 2020

©2020. The Authors

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Grounding Line Retreat of Denman Glacier, East Antarctica, Measured With COSMO-SkyMed Radar Interferometry Data

V. Brancato¹, E. Rignot^{1,2,3}, P. Milillo², M. Morlighem², J. Mouginot^{2,4}, L. An², B. Scheuchl², S. Jeong², P. Rizzoli⁵, J. L. Bueso Bello⁵, and P. Prats-Iraola⁵

¹Jet Propulsion Laboratory, California Institute of Technology, Pasadena, USA, ²Department of Earth System Science, University of California, Irvine, USA, ³Department of Civil and Environmental Engineering, University of California, Irvine, USA, ⁴University Grenoble Alpes, CNRS, INP, Grenoble IGE, Grenoble, France, ⁵German Aerospace Center (DLR), Microwaves and Radar Institute, Weßling, Germany

Abstract Denman Glacier, East Antarctica, holds an ice volume equivalent to a 1.5 m rise in global sea level. Using satellite radar interferometry from the COSMO-SkyMed constellation, we detect a 5.4 ± 0.3 km grounding line retreat between 1996 and 2017–2018. A novel reconstruction of the glacier bed topography indicates that the retreat proceeds on the western flank along a previously unknown 5 km wide, 1,800 m deep trough, deepening to 3,400 m below sea level. On the eastern flank, the grounding line is stabilized by a 10 km wide ridge. At tidal frequencies, the grounding line extends over a several kilometer-wide grounding zone, enabling warm ocean water to melt ice at critical locations for glacier stability. If warm, modified Circumpolar Deep Water reaches the sub-ice-shelf cavity and continues to melt ice at a rate exceeding balance conditions, the potential exists for Denman Glacier to retreat irreversibly into the deepest, marine-based basin in Antarctica.

Plain Language Summary Using satellite radar data from the Italian COSMO-SkyMed constellation, we document the grounding line retreat of Denman Glacier, a major glacier in East Antarctica that holds an ice volume equivalent to a 1.5 m global sea level rise. The grounding line is retreating asymmetrically. On the eastern flank, the glacier is protected by a subglacial ridge. On the western flank, we find a deep and steep trough with a bed slope that makes the glacier conducive to rapid retreat. If warm water continues to induce high rates of ice melt near the glacier grounding zone, the potential exists for Denman Glacier to undergo a rapid and irreversible retreat, with major consequences for sea level rise.

1. Introduction

Over the last decades, the Antarctic Ice Sheet has been contributing to global sea level rise, and the marine-based Wilkes Land sector of East Antarctica has been an important contributor to the total mass loss (Rignot et al., 2019). Denman Glacier (DG), one of the largest glaciers in East Antarctica in terms of ice discharge, holds an ice volume equivalent to a 1.5 m global sea level rise (Rignot et al., 2019). Radar altimetry data from the European Envisat mission indicated that DG thinned at 0.4 m/year between 2002 and 2010 upstream of its grounding line (Flament & Rémy, 2012), where the glacier speed reaches 1.4 km/year (Young et al., 1989). DG has experienced a cumulative mass loss of 268 ± 19 Gt between 1979 and 2017, or 7.0 ± 0.5 Gt/year, or $13.2 \pm 1\%$ of its balance flux of 52.6 ± 4.2 Gt/year (Rignot et al., 2019) (Figure S1 in the supporting information). The 24,000 km² floating extension of the glacier, which includes Shackleton Ice Shelf and Denman Ice Tongue (DIT), has been melting at an area-average rate of 3.1 ± 0.7 m/year, which is above average among other ice shelves in East Antarctica (Rignot et al., 2013).

The first and only delineation of the grounding line of DG was derived using Earth Remote Sensing (ERS-1/2) satellite data from year 1996 (Rignot, 2002). The bathymetry in front of and beneath DIT is not well known. Seismic traverses and airborne surveys over land have revealed the presence of a deep trough, more than 2,000 m below sea level, 100 km upstream of the 1996 grounding line (Young et al., 1989). The trough connects with the deep marine Aurora Subglacial Basin, about 400 km to the southeast (Young et al., 2011). There is no detail about the depth of DG near its grounding line.

Here, we present a new delineation of DG grounding line using a time series of Differential Interferometric Synthetic Aperture Radar (DInSAR) data from years 2016 to 2018. To quantify the thinning rate of the glacier and the melt rate of its floating ice shelf, we use a time series of high resolution Digital Elevation Models (DEMs) of the glacier surface topography from the TanDEM-X (TDX) radar mission and laser altimetry data from NASA's Operation IceBridge (OIB) mission. We present a new bed topography of the glacier that combines OIB ice thickness and vector ice velocity data on land. The map is extended offshore using a three-dimensional inversion of gravity data to reveal the bathymetry in front and beneath the ice shelf. We evaluate the short-term variability in grounding line position using the 2016–2018 data and its long-term variability in reference to the ERS-1/2 data from 1996. We compare the results with the newly reconstructed bed topography and bathymetry to conclude on the recent and future evolution of DG.

2. Data and Methods

2.1. Grounding Line Mapping

To delineate the grounding line of DG, we form SAR interferograms with a 1-day time interval by coherently combining SAR acquisitions from the ERS-1/2 satellites in year 1996 and from the Agenzia Spaziale Italiana's COSMO-SkyMed (CSK) constellation in years 2016, 2017, and 2018 (Table S1). We multilook each CSK SAR interferogram using an 8×8 averaging window in range and azimuth, equivalent to $24 \text{ m} \times 24 \text{ m}$ on the ground and use a TDX DEM at 30 m spacing to remove the topographic signal (Milillo et al., 2017; Rignot, 2002). We form 24 CSK differential pairs (DINSAR) by differencing consecutive 1-day CSK SAR interferograms (Figure S2). The differencing eliminates the horizontal motion of the glacier to reveal its differential motion, which includes the vertical motion of floating ice in response to changes in oceanic tides (Brunt et al., 2011; Rignot et al., 2011), changes in surface elevation caused by subglacial lake drainage or filling on land (Gray et al., 2005), and residual data noise (e.g., turbulent water vapor and thermal noise). We reprocess the two ERS DINSAR interferograms from year 1996 improving the interferometric baseline determination using OIB altimetry data and the topographic correction using a TDX DEM.

The differential signal associated with tidal motion is characterized by a step function immediately seaward of the grounding line, over a region typically 5–9 km wide called the flexure zone, where ice adjusts to hydrostatic equilibrium (Brunt et al., 2011). In the flexure zone, the deformation signal results from a quadruple difference in tidal state between the four epochs used to form the DInSAR interferogram (Rignot et al., 2011). The small amplitude of the step function makes arduous the delineation of the grounding line. Therefore, we use only DInSAR data with a strong differential tidal signal, that is, typically more than 10 fringes (or 360° variation in interferometric phase) across the flexure zone (Figure S2). The grounding line is identified manually at the inward limit of the vertical displacement zone of the ice surface. The most upstream interferometric fringe represents the farthest point inland where seawater intruded beneath the glacier among the four different tidal epochs and modified its ice surface elevation (Figure S2) (Milillo et al., 2017; Rignot, 2002).

The tidal signal is easily separated from the signal associated with the drainage/filling of subglacial lakes, which has a circular to oval shape, a different signal amplitude than the tidal signal, and occurs upstream of the grounding line. Conversely, the number of fringes between the grounded ice (inland area with no residual fringes) and freely floating ice (seaward of the zone of tidal flexure) must always be the same in each DInSAR interferogram, that is, equal to the difference in tidal height between the four different epochs projected along the radar line of sight. This eliminates the possibility of placing the grounding line on the wrong set of fringes. This guideline is crucial to correctly map the grounding line across the entire glacier since the intrusion of seawater beneath grounded ice varies spatially across the grounding line as a function of tidal state, bed slope, and ice thickness. In some places, for example, the western flank of DG, the tidal deformation extends farther inland than on average. The fringe spacing in the flexure zone increases relative to the average fringe spacing elsewhere, yet the number of tidal fringes across the flexure zone remains the same, which helps reduce the risk of misidentifying the grounding line. For each DInSAR data, we digitize grounding line segments with an uncertainty in grounding line position based on the amplitude of the tidal signal and phase noise.

The CSK data do not provide a complete coverage of the grounding line of Scott and Northcliffe glaciers. For those glaciers, we complement the CSK data with a 6-day DInSAR interferogram from the European Space Agency (ESA) Sentinel-1A/B (S1-A/B) in year 2017 processed as reported in Scheuchl et al. (2016)

(Figure S3). However, the S1-A/B DInSAR data cannot be used to detect the grounding line of DG since the repeat cycle of S1-A/B is too long to maintain phase coherence over DG.

To reconstruct the tidal state at the time of passage of the satellites (i.e., 4 epochs for ERS-1/2, 4 epochs for CSK), we use the CircumAntarctic Tide Simulation (CATS2008A) model at a center location of $64^{\circ}43.2831'$ S, 100°0.5712'E, just off the DIT, in the open ocean (Padman et al., 2002). We assume that the floating ice responds elastically to changes in oceanic tides. To evaluate the tidal model, we compare the differential tidal signal measured from the DInSAR data (i.e., step function in height between grounded and freely floating ice) with the tidal differential height estimated from the tidal model (Figure S4). For each DInSAR data, we calculate the maximum amplitude of the tidal signal among the four tidal states used to form the DInSAR interferogram, or h_{max} .

2.2. Ice Surface and Bed Elevation

We generate a time series of DEMs at 12 m spacing, with meter-scale vertical precision, using bistatic SAR acquisitions from the TDX mission and a standard processing chain described in Rizzoli et al. (2017). We use data from the Ice Cloud and land Elevation Satellite (ICESat) mission, (GLAH14, Version 34; Zwally et al., 2014) and OIB (i.e., Riegl laser altimeter; Blankenship et al., 2012) to calibrate the TDX DEMs absolute height over rocky outcrops (Milillo et al., 2019). We estimate the height precision and absolute height accuracy of the TDX DEMs to be better than 2 m (Rizzoli et al., 2017).

We employ a new reconstruction of DG bed topography from BedMachine Antarctica using a mass conservation approach, with a grid spacing of 500 m and a nominal accuracy of 100 m (Morlighem et al., 2019). The mass conservation method over the study region combines ice surface velocity vectors from years 2007–2008 (Mouginot et al., 2017), ice thickness data from the HiCARS sounder (Hi-CApability Radar Sounder) (Young et al., 2011), surface elevation from the TDX DEM from year 2017, and surface mass balance averaged for the years 1979 to 2008 from Regional Atmospheric Climate Model Version 2.3p1 (RACMO2.3p1) (Rignot et al., 2019; van Wessem et al., 2014). In this region, RACMO2.3p1 is superior to RACMO2.3p2 (Mohajerani et al., 2018, 2019). Temporal changes in ice thickness (≤ 1 m/year) and basal melt beneath grounded ice (\leq cm/year) are negligible. Full radar sounder-derived thickness crossings of DG are only available 120 km upstream of the grounding line.

2.3. Elevation Changes and Ice Shelf Melt

We quantify changes in surface elevation by differencing TDX DEMs separated by 2 to 3 years using both an Eulerian (i.e., fixed reference grid) and a Lagrangian (i.e., reference grid moving with the ice) approach. These methods provide distinct information over the grounded and floating portion of DG. In an Eulerian framework, changes in surface elevation over grounded ice reflect variation in ice thickness, *H*, caused by ice dynamics and snowfall accumulation since the elevation of the ice base does not change (Shean et al., 2019). Under the assumption of incompressibility and assuming a column-average velocity **v**, mass conservation applied to an ice column of thickness *H* dictates $dH/dt = -(\nabla \cdot H\mathbf{v}) + \dot{a} - \dot{b}$, where ∇ is the divergence operator, \dot{a} is the surface accumulation rate, and \dot{b} is the basal melt rate. Due to the advection of heterogeneities in ice thickness, this methodology is not useful on floating ice. To calibrate dH/dt, we use ice-free areas near Burger Hills where changes in surface elevation should be zero. We smooth the results using a 15 × 15 pixels (420 m × 420 m) moving window and grid the results at 1.5 km spacing. We estimate a mean error of 0.05 m/year and a standard deviation less than 0.1 m/year.

On floating ice, temporal changes in surface elevation obtained in a Lagrangian framework (Moholdt et al., 2014; Shean et al., 2019) yield estimates of ice shelf melt rates assuming a uniform rate of horizontal motion and ice to be in hydrostatic equilibrium as $Dh/dt + h(\nabla \cdot \mathbf{v}) = (a - b)(\rho_i - \rho_{sw})/(\rho_{sw})$, where Dh/dt is the material derivative, ∇ the gradient operator, \mathbf{v} the vector velocity, \dot{a} the surface mass balance, m_b the ocean-induced ice shelf melt rate with positive (negative) values indicating melting (freezing), $\rho_i = 0.917$ g/cm³ is the density of ice, and $\rho_{sw} = 1.028$ g/cm³ is the density of sea water. We use a chip size of 30 × 30 pixels (840 m × 840 m) with step size of 16 pixels (448 m) in both directions and smooth the results with a 15 × 15 pixels (420 m × 420 m) moving average window to obtain melt rates at 1.5 km spacing. We follow the trajectory of each pixel (on a monthly basis) using the velocity map and correct for tidal motion at the time of the satellite overflights using the CATS2008A model (Table S2).

The main sources of uncertainty of the estimation of ice shelf melt are the measurement errors (e.g., DEM, ice velocity), correction of ocean tides, changes in surface elevation due to snowfall accumulation, and ice

dynamic thinning. We assume these error sources to be unbiased. We assume uncorrelated error sources for the TDX surface elevation (2 m) (Rizzoli et al., 2017) and yearly-averaged ice velocity maps (2–3 m/year) (Mouginot et al., 2017). The error in tide correction is 6 cm. We use a depth-averaged density for ice and ocean water, both assumed spatially and temporally constant. We extract the surface component of the transient melt from RACMO2.3p1 model (van Wessem et al., 2014), which is less than 1 m/year, yielding an uncertainty of few cm per year. These results must be interpreted with caution over grounded ice and in the transition zone between grounded and floating ice wherein the assumption of hydrostatic equilibrium is not strictly valid due to bending stresses of ice and adjustments of ice to floatation over several km. We correct our estimates of ice shelf melt rate for dynamic thinning (0.4 m/year) measured on grounded ice, assuming that the same dynamic thinning applies to floating ice. The corresponding error is 0.1 m/year in height or 1 m/year in ice thickness. Considering all the error sources, the total uncertainty of the melt rate estimates is 4 m/year.

2.4. Oceanographic data

We use hydrographic data collected by instrumented marine mammals in 2004–2005 and 2011–2015 distributed by the Marine Mammals Exploring the Oceans Pole to Pole (MEOP) consortium (Treasure et al., 2017). Each mammal is equipped with a Conductivity Temperature Density Satellite Relay Data Logger (CTD-SRDL). Before deployment, each sensor is calibrated with a precision of 0.005 °C and 0.005 mS/cm, respectively, for temperature and conductivity. Tags recovered after deployments (i.e., 72–136 days) show no change in calibration within the manufacturer's specification (Roquet et al., 2014; Treasure et al., 2017). Each time the mammal surfaces, the data collected throughout a dive are processed by the CTD-SRDL sensors and telemetered through the Argo satellite system. The geolocation of each dive is determined by the Argo satellite triangulation with an accuracy of 4 km. To ensure data quality, the hydrographic data are postprocessed as in Roquet et al. (2014). We find many CTD casts near the edge of the continental shelf but few in the vicinity of DIT (Figures S5 and S6; Table S3).

2.5. Bathymetry

The International Bathymetric Chart of the Southern Ocean (IBCSO) (Arndt et al., 2013) includes no depth data on the continental shelf and beneath DIT. The water column beneath DIT was arbitrarily set at 40 m. The ice shelf thickness was inferred from a DEM of Antarctica assuming ice in hydrostatic equilibrium. Here, we employ a three-dimensional inversion of airborne gravity data to infer the bathymetry beneath DIT using constraints from BedMachine Antarctica on land (Morlighem et al., 2019), CTD maximum depth data at sea, islands, and the presence of ice rises (zero water column) (Figure S6). To invert the gravity data, we use the Geosoft GM-SYS 3-D software, which uses the Parker method (Parker, 1973). This method calculates the gravity anomaly caused by an uneven, uniform layer of material by means of a series of Fourier transforms and modifies it to minimize the misfit between calculated and observed gravity. We model the domain as three horizontal layers: (1) a solid ice layer with a density of 0.917 g/cm³; (2) an ocean water layer with a density of 1.028 g/cm³; and (3) a rock layer with a uniform density of 2.67 g/cm³ following An et al. (2019). Airborne gravity data are from OIB, Investigating the Cryospheric Evolution of the Central Antarctic Plate (ICECAP) (Young et al., 2011) and from the Gravity and Geoid in Antarctica (AntGG) international compilation (Scheinert et al., 2016) (Figure S7).

We run a forward model of the gravity field using an initial bed solution. In areas with bed elevation constraints, we calculate the DC shift (Direct Current) or mean difference between modeled and observed gravity. We then interpolate the DC shift over the entire domain using a minimum curvature algorithm (An et al., 2019). The DC shift captures the long wavelength (100 km) variations in geology below the model domain and removes it from the observations. As a result of the inversion, the misfit between modeled and observed gravity is reduced from 15.5 mGal with the initial bed solution to 5.8 mGal with the final bed solution. We estimate that the nominal precision of the bathymetry to be 150 to 200 m, but local errors larger than 200 m cannot be excluded (An et al., 2019).

3. Results

The availability of multiple grounding line measurements with different tidal states on the same calendar year allows to identify the grounding zone (Figures 1, 2, and S8) of DG, that is, the region where the grounding line migrates back and forth with changes in oceanic tides (Milillo et al., 2017; Sayag & Worster, 2011) and also to determine the average rate of grounding line retreat with more confidence. The grounding line



Figure 1. (a) Denman Glacier (DG), East Antarctica (red box). (b) 2017 ice velocity flow lines color-coded on a logarithm scale with the 1996 grounding line (black) and the 2019 ice front (red). Ice flows from left (grounded ice) to right (floating ice). At sea, ocean temperature from Southern Ocean State Estimate (SOSE) (Mazloff et al., 2010) at 310 m depth color-coded between -2 to +2 °C. Color-coded rectangular markers represent MEOP-CTD in situ temperatures (Roquet et al., 2014) at the maximum depth of the cast. (c) At sea, bathymetry from a three-dimensional gravity inversion (this study) and IBCSO (Arndt et al., 2013) with 100 m contours in black. On land, bed topography from mass conservation. Inset in (d) shows details at the grounding line (2017: green; 2018: yellow). (e) Bed elevation beneath DG and Denman Ice Tongue (DIT) along profile (f) A-A' (purple dashed line) with bed (brown), ice (light blue), and ocean (dark blue).



Figure 2. (a) Grounding zone of Denman Glacier in 1996 (black), 2016 (violet), 2017 (green), and 2018 (yellow) overlaid on a bed topography from mass conservation with 100 m contours in black. Grounding line position versus maximum tidal height, h_{max}, at the times of passages of the satellites for different years along the western flank (b) and eastern flank (c) of Denman Glacier (DG), East Antarctica. Positive values of the grounding line position with respect to the reference points A and B indicate that the grounding line retreated inland. The horizontal error bar is the uncertainty in tidal state. The vertical error bar is the uncertainty in grounding line delineation in the DInSAR interferograms. Labeling of the individual points refers to the labeling of the DInSAR data shown in Figure S2.

is expected to be at its most retreated position at high tides, when ice is lifted off the bed by rising sea levels, and at its most advanced position at low tides, when ice is pushed down by receding sea levels. We quantify the grounding line migration on the western (Figure 2b) and eastern flanks (Figure 2c) by computing the distance between the reference grounding line in 1996 and the grounding lines in 2016–2018 along the glacier flow lines at two selected locations in the deepest parts of the glacier troughs (Figure 2).

The error weighted mean of the grounding line retreat between 1996 and 2016–2018 is 5.7 ± 0.3 km. Yet the 2017 and 2018 DInSAR data include more observations with high h_{max} values, which favor more retreated positions (Figure 2). Therefore, we calculate a weighted mean for the DInSAR data where h_{max} is within of 15 cm of neutral tide to obtain a grounding line retreat of 5.4 ± 0.3 km on the western flank and 0.35 ± 0.1 km on the eastern flank. The range (min-max) is 1.3 to 9.9 km for the west flank and -0.3 to 5.7 km for the eastern flank.

On neighboring glaciers, we find no retreat on Scott Glacier, a 2 to 3 km retreat on Reid Glacier, and no conclusion possible for Northcliffe Glacier, which is incompletely covered by CSK data.

The reconstructed bed topography (Figure 1c) reveals a 10 km wide subglacial ridge, 400 to 700 m below sea level, 4 km upstream of the present-day grounding line on the eastern flank, and a 5 km wide, 1,800 m below sea level trough on the west flank, with no ridge. The western trough deepens for another 18 km in the inland direction to 1,900 m below sea level, before widening to a total width of 12 km and deepening further to 3,400 m below sea level over the next 30 km inland, that is, a steep retrograde bed slope of 5%. This maximum depth makes this region one the deepest locations in Antarctica (Fretwell et al., 2013; Young et al., 2011). Beyond 50 km inland, the bed elevation rises but remains well below sea level (Figure 1).

From the Eulerian approach, we find that the main trunk of DG thinned at 0.4 ± 0.1 m/year between 2011 and 2013, with areas surrounding the glacier thickening at 0.35 ± 0.1 m/year (Figure 3c). This magnitude thinning is consistent with prior altimetry records (Flament & Rémy, 2012; McMillan et al., 2018) where thinning was also confined in the areas of fast flow. This signal suggests that ice flows faster than the speed required to maintain a state of mass balance. Indeed, DG has been speeding up by $16 \pm 2\%$ since the 1970s and is estimated to be $10 \pm 2\%$ out of balance at present (Rignot et al., 2019). Most glacier thinning is confined in the last 50–100 km of grounded ice, coincident with the area of speed up (Figure 3b). On DIT, the speed up was larger, 33\% between 1957 and 1996, and another 10\% until 2016 (Dolgushin, 1965; Rignot et al., 2019), implying that the ice shelf is thinning dynamically as well.

From the Lagrangian approach (Moholdt et al., 2014; Shean et al., 2019), we estimate ice shelf melt rates in excess of 45 ± 4 m/year near the grounding line of DG between 2011 and 2014. Within the grounding zone,

Geophysical Research Letters



Figure 3. (a) 2017 CSK DInSAR observation of tidal motion with grounding lines from 1996 (black), 2017 (green), and 2018 (yellow) overlaid on Moderate Resolution Imaging Spectroradiometer (MODIS) mosaic (Haran et al., 2014). (b) Change in ice speed between 1995 and 2017 color-coded from red (acceleration) to blue (deceleration) with 50 m/year contours. (c) TDX surface elevation change between 2011 and 2013 color-coded from red (thinning) to blue (thickening) with 100 m surface elevation contours. (d) Ice shelf melt rate between 2011 and 2014 color-coded from blue (freeze on) to red (melt) and purple (rapid melt) with 5 m/year contours. Insets in a-d reveal details at the grounding line.

the methodology is affected by uncertainties in ice floatation (Wilson et al., 2017), but our results indicate melt rates in the range of 10–20 m/year. Such ice shelf melt rates near the grounding line are high compared to other East Antarctic ice shelves (Rignot et al., 2013), but comparable in magnitude to those observed in warm parts of West Antarctica, for example, the Amundsen Sea Embayment (ASE). In the grounding zone, a narrow region that is not always in contact with seawater, there are no prior estimates of ice shelf melt rates, but the rates we estimate are significant.

The 45 m/year melt on the ice shelf requires the presence of relatively warm water, typically in the range of 4 to 5 °C above the in situ freezing point of seawater (Rignot & Jacobs, 2002). The five MEOP casts from year 2011 near DIT reveal in situ ocean temperatures ranging from -1.31 to -0.26 °C, at depths of 550 to 850 m, which correspond to a thermal forcing (difference between the in situ temperature and the depth-dependent, salinity-dependent freezing point of seawater) of 2.6 to 3.2 °C at 1,000 m depth (Fujino et al., 1974).

The new reconstructed bathymetry reveals a sinuous, deep trough beneath DIT intersected by three ridges at about 500 m depth (Figure 1f). The first ridge, approximately 60 km north of the grounding line, yields a water column thickness (difference between the seafloor depth and the depth of the ice shelf draft) less than 200 m. At this location, the DInSAR data do not suggest the presence of any ephemeral grounding points, that is, locations at which the ice shelf touches the sea floor at low tides (Schmeltz et al., 2002). Therefore, it is likely that warm water, possibly of Circumpolar Deep origin, circulates freely beneath the shelf (more than 400 m deep) and reaches DG grounding line. The two ridges farther north may also affect the heat transfer to the cavity, but the water column in that location is thicker since the ice shelf is thinner. To quantify the heat transfer, it would be essential to better constrain the depth of the ridge and obtain observations of the water structure in the ice shelf cavity.

4. Discussion

Our analysis of the DInSAR data spanning from 1996 to 2018 indicates that the grounding line of DG has been retreating asymmetrically, with a maximum retreat on the western flank along a previously unknown deep trough with a retrograde slope, and almost no retreat on the eastern flank, which is blocked by a prominent ridge. The 5 to 6 km retreat in 21 years along the western flank is consistent with the presence of a retrograde bed and high ice shelf melt rates along a grounding line more than 2 km below sea level. To the east, Scott Glacier is thinning upstream of its grounding line, but the grounding line is stabilized on a ridge. We do not have enough data on Northcliffe Glacier to quantify its retreat, but its grounding zone was exceptionally wide (\geq 10 km) in 1996, which suggests the presence of a nearly flat bed at the grounding line (Sayag & Worster, 2011). Since the glacier does not extend below sea level far inland, it does not hold much potential for sea level rise. Similarly, Reid Glacier is retreating but does not drain a large marine-based sector.

On the west flank of DG, the grounding zone is several km wide, that is, seawater infiltrate beneath grounded ice over considerable distances at high tide (Brunt et al., 2011). This wide grounding zone cannot be explained by the interaction of solid ice with a rigid bed but can be explained by the presence of a viscoelastic bed as in Sayag and Worster (2011). One important consequence is that ice melts in contact with ocean waters over considerable distances in a region that exerts a critical control on glacier stability (Sayag & Worster, 2013). If warm water present on the continental shelf reaches the deeper part of DG sub-ice-shelf cavity, the thermal forcing increases to 4.2 °C, in principle sufficient to justify an average melt rate of 45 m/year (Rignot & Jacobs, 2002). Vigorous ocean-induced ice melt within the grounding zone will directly reduce basal resistance to flow and entrain glacier speed up and further retreat. Such type of ice-ocean interaction, which is not included in current ice sheet models (Nowicki et al., 2016; Seroussi & Morlighem, 2018), could significantly increase projections of sea level rise from this region.

To the best of our knowledge, the bed slope configuration near the grounding line of DG is unique in this sector of East Antarctica. Totten Glacier, which is retreating at a comparable rate, must retreat for 50 km along prograde bed slopes before reaching retrograde bed slopes (Li et al., 2015). Moscow University Glacier is also flanked by prograde slopes that should provide more stability (Morlighem et al., 2019).

The grounding line retreat of DG is one order of magnitude smaller than that observed in the ASE (Scheuchl et al., 2016), However, the sector containing DG holds an ice volume equivalent larger than for the ASE sector (1.2 m); hence, any signal of retreat along DG is of considerable importance for the future. The new bed topography data suggest the presence of a direct retrograde pathway for the glacier, that is, no bump in

bed topography above the current grounding line that could slow down the retreat. One factor that could slow the retreat is the narrowness of the deep channel, which may provide significant lateral drag, similar to Jakobshavn Isbrae in Greenland which retreats at 0.5 km/year (Bondzio et al., 2017; Habermann et al., 2013; Thomas, 2004). A second factor would be the efficiency of ocean heat transfer to the DG grounding line. In the future, it will be critical to collect oceanographic data in that region and continue observations of the grounding line dynamics, which is currently only possible with the CSK constellation.

5. Conclusions

A detailed mapping of the grounding line of Denman Glacier, East Antarctica, with satellite radar interferometry, reveals a complex and asymmetric pattern of retreat. By comparing satellite data acquired 21 years apart, we detect a 5.4 ± 0.3 km retreat on the western flank, proceeding along a previously unknown, smooth, retrograde bed dropping to more than 3,400 m depth over the next 50 km. The retreat is negligible on the eastern part of the glacier, which is protected by a subglacial ridge. Existing ocean data and observations of ice shelf melt rates suggest the presence of vigorous ice-ocean interaction beneath the Denman Ice Tongue, which must play a central role in the current evolution of the glacier, similar to what is observed in the ASE sector of West Antarctica. We conclude that the potential exists for a rapid retreat along the Denman trough in the future. A strengthening of the westerlies (Spence et al., 2014), caused by rapid midlatitude climate warming and cooling of the stratosphere by the ozone depletion (Swart et al., 2018), may bring more warm Circumpolar Deep Water toward the coast of Antarctica and explain the present-day evolution of this sector. These observations challenge the view of glacier stability in East Antarctica.

References

An, L., Rignot, E., Millan, R., Tinto, K. J., & Willis, J. (2019). Bathymetry of Northwest Greenland using "Ocean Melting Greenland" (OMG) high-resolution airborne gravity and other data. *Remote Sensing*, 11, 131–147.

- Arndt, J. E., Schenke, H. W., Jakobsson, M., Nitsche, F., Buys, G., Goleby, B., & Wigley, R. (2013). The International Bathymetric Chart of the Southern Ocean (IBCSO) Version 1.0—A new bathymetric compilation covering circum-Antarctic waters. *Geophysical Research Letters*, 40, 3111–3117. https://doi.org/10.1002/grl.50413
- Blankenship, D. D., Kempf, S. D., & Young, D. A. (2012). IceBridge Riegl laser altimetry L1B time-tagged laser ranges, Version 1. https:// doi.org/10.5067/KH078MNXPAOV
- Bondzio, J. H., Morlighem, M., Seroussi, H., Kleiner, T., Ruckamp, M., Mouginot, J., & Humbert, A. (2017). The mechanisms behind Jakobshavn Isbræ's acceleration and mass loss: A 3D thermomechanical model study. *Geophysical Research Letters*, 44, 6252–6260. https:// doi.org/10.1002/2017GL073309
- Brunt, K. M., Fricker, H. A., & Padman, L. (2011). Analysis of ice plains of the Filchner-Ronne Ice Shelf, Antarctica, using ICESat laser altimetry. *Journal of Glaciology*, 57, 965–975.
- Dolgushin, L. D. (1965). New data on the rates of movement of Antarctic glaciers. *Information bulletin of the Soviet Antarctic expedition*, 56, 17–20.
- Flament, T., & Rémy, F. (2012). Dynamic thinning of Antarctic glaciers from along-track repeat radar altimetry. Journal of Glaciology, 58, 830–840.
- Fretwell, P., Pritchard, H. D., Vaughan, D. G., Bamber, J. L., Barrand, N. E., Bell, R., & Zirizotti, A. (2013). Bedmap2: Improved ice bed, surface and thickness datasets for Antarctica. *The Cryosphere*, 7(1), 375–393.
- Fujino, K., Lewis, E. L., & Perkin, R. G. (1974). The freezing point of seawater at pressures up to 100 bars. Journal of Geophysical Research, 79, 1792–1797.
- Gray, L., Joughin, I., Tulaczyk, S., Spikes, V. B., Bindschadler, R., & Jekek, K. (2005). Evidence for subglacial water transport in the West Antarctic Ice Sheet through three-dimensional satellite radar interferometry. *Geophysical Research Letters*, 32, L03501. https://doi.org/ 10.1029/2004GL021387
- Habermann, M., Truffer, M., & Maxwell, D. (2013). Changing basal conditions during the speed-up of Jakobshavn Isbræ, Greenland. *The Cryosphere*, 7, 1679–1692.
- Haran, T., Bohlander, J., Scambos, T., Painter, T., & Fahnestock, M. (2014). MODIS Mosaic of Antarctica 2008-2009 (MOA2009) Image Map, Version 1, Remote Sensing of Environment. Boulder, Colorado USA: NSIDC: National Snow and Ice Data Center. https://doi.org/ 10.7265/N5KP8037
- Li, X., Rignot, E., Morlighem, M., Mouginot, J., & Scheuchl, B. (2015). Grounding line retreat of Totten Glacier, East Antarctica, 1996 to 2013. Geophysical Research Letters, 42, 8049–8056. https://doi.org/10.1002/2015GL065701

Mazloff, M., Heimbach, P., & Wunsch, K. (2010). An eddy-permitting Southern Ocean State Estimate. *Journal of Physical Oceanography*, 40, 880–899.

- McMillan, M., Shepherd, A., Muir, A., Gaudelli, J., Hogg, A. E., & Cullen, R. (2018). Assessment of CryoSat-2 interferometric and non-interferometric SAR altimetry over ice sheets. *Advances in Space Research*, *62*, 1281–1291.
- Milillo, P., Rignot, E., Mouginot, J., Scheuchl, B., Morlighem, M., & Salzer, J. T. (2017). On the short-term grounding zone dynamics of Pine Island Glacier, West Antarctica, observed with COSMO-SkyMed interferometric data. *Geophysical Research Letters*, 44, 10,436–10,444. https://doi.org/10.1002/2017GL074320
- Milillo, P., Rignot, E., Rizzoli, P., Scheuchl, B., Mouginot, J., Bueso Bello, J. L., & Prats-Iraola, P. (2019). Heterogeneous retreat and ice melt of Thwaites Glacier, West Antarctica. *Science Advances*, 5, eaau3433.
- Mohajerani, Y., Velicogna, I., & Rignot, E. (2018). Mass loss of Totten and Moscow University Glaciers, East Antarctica, using regionally optimized GRACE mascons. *Geophysical Research Letters*, 45, 7010–7018. https://doi.org/10.1029/2018GL078173

Mohajerani, Y., Velicogna, I., & Rignot, E. (2019). Evaluation of regional climate models using regionally optimized GRACE Mascons in the Amery and Getz Ice Shelves Basins, Antarctica. *Geophysical Research Letters*, 46, 13,883–13,891. https://doi.org/10.1029/2019GL084665

Acknowledgments

This work was conducted at the Jet Propulsion Laboratory, California Institute of Technology, and at the UC Irvine under a contract with the Cryosphere Program of the National Aeronautics and Space Administration, ©2019, California Institute of Technology. All rights reserved. We thank the Italian Space Agency (ASI) for providing CSK data (original COSMO-SkyMed product ©ASI. Agenzia Spaziale Italiana (2008-2017)) and the German Space Agency (DLR) for providing TDX data (original TanDEM-X product ©DLR (2007-2017)). The marine mammal data were collected and made freely available by the International MEOP Consortium and the national programs that contribute to it (https://www. meop.net). Displacement fields, SAR processed data, bathymetry, and bed topography are available online at https://doi.org/10.15146/zf0j-5m50.



Moholdt, G., Padman, L., & Fricker, H. A. (2014). Basal mass budget of Ross and Filchner-Ronne ice shelves, Antarctica, derived from Lagrangian analysis of ICESat altimetry: Ice shelf basal melting from altimetry. *Journal of Geophysical Research: Earth Surface*, 119, 2361–2380. https://doi.org/10.1002/2014JF003171

Morlighem, M., Rignot, E., Binder, T., Blankenship, D., Drews, R., Eagles, G., & Young, D. A. (2019). Deep glacial troughs and stabilizing ridges unveiled beneath the margins of the Antarctic ice sheet. *Nature Geoscience*, 13, 132–137. https://doi.org/10.1038/ s41561-019-0510-8

Mouginot, J., Rignot, E., Scheuchl, B., & Millain, R. (2017). Comprehensive annual ice sheet velocity mapping using Landsat-8, Sentinel-1, and RADARSAT-2 data. *Remote Sensing*, 9, 364–384.

Nowicki, S. M. J., Payne, A., Larour, E., Seroussi, H., Goelzer, H., Lipscomb, W., & Shepherd, A. (2016). Ice Sheet Model Intercomparison Project (ISMIP6) contribution to CMIP6. Geoscientific Model Development, 12, 4521–4545.

Padman, L., Fricker, H. A., Coleman, R., Howard, S., & Erofeeva, L. (2002). A new tide model for the Antarctic ice shelves and seas. Annals of Glaciology, 34, 247–254.

Parker, R. L. (1973). The rapid calculation of potential anomalies. Geophysical Journal of the Royal Astronomical Society, 31, 447–455.

Rignot, E. (2002). Mass balance of East Antarctic glaciers and ice shelves from satellite data. Annals of Glaciology, 34, 217–227.

Rignot, E., & Jacobs, S. (2002). Rapid bottom melting widespread near Antarctic Ice Sheet grounding lines. Science, 296, 2020–2023.

Rignot, E., Jacobs, S., Mouginot, J., & Scheuchl, B. (2013). Ice-shelf melting around Antarctica. Science, 341, 266-270.

Rignot, E., Mouginot, J., & Scheuchl, B. (2011). Antarctic grounding line mapping from differential satellite radar interferometry. *Geophysical Research Letters*, 38, L10504. https://doi.org/10.1029/2011GL047109

Rignot, E., Mouginot, J., Scheuchl, B., van den Broeke, M. J., & Morlighem, M. (2019). Four decades of Antarctic ice sheet mass balance from 1979 to 2017. Proceedings of the National Academy of Sciences of the United States of America, 116, 109–1103.

Rizzoli, P., Martone, M., Gonzalez, C., Wecklich, C., Tridon, D. B., Bräutigam, B., & Moreira, A. (2017). Generation and performance assessment of the global TanDEM-X digital elevation model. ISPRS Journal of Photogrammetry and Remote Sensing, 132, 119–139.

Roquet, F., Williams, G. D., Hindell, M. A., Harcourt, R., McMahon, C., Guinet, C., & Fedak, M. (2014). A Southern Indian Ocean database of hydrographic profiles obtained with instrumented elephant seals. *Scientific Data*, *1*, 140,028.

Sayag, R., & Worster, M. G. (2011). Elastic response of a grounded ice sheet coupled to a floating ice shelf. *Physical Review E*, 84, 36,111.

- Sayag, R., & Worster, M. G. (2013). Elastic dynamics and tidal migration of grounding lines modify subglacial lubrication and melting. Geophysical Research Letters, 40, 5877–5881. https://doi.org/10.1002/2013GL057942
- Scheinert, M., Ferraccioli, F., Schwabe, J., Bell, R., Studinger, M., Damaske, D., & Richter, T. D. (2016). New Antarctic gravity anomaly grid for enhanced geodetic and geophysical studies in Antarctica. *Geophysical Research Letters*, 43, 600–610. https://doi.org/10.1002/ 2015GL067439
- Scheuchl, B., Mouginot, J., Rignot, E., Morlighem, M., & Khazendar, A. (2016). Grounding line retreat of Pope, Smith, and Kohler Glaciers, West Antarctica, measured with Sentinel-1a radar interferometry data. *Geophysical Research Letters*, 43, 8572–8579. https://doi.org/10. 1002/2016GL069287

Schmeltz, M., Rignot, E., Dupont, T. D., & Mac Ayeal, D. R. (2002). Sensitivity of Pine Island Glacier, West Antarctica, to changes in ice-shelf and basal conditions: A model study. Journal of Glaciology, 48, 552–558.

- Seroussi, H., & Morlighem, M. (2018). Representation of basal melting at the grounding line in ice flow models. *The Cryosphere*, 12, 3085–3096.
- Shean, D., Joughin, I. R., Dutrieux, P., Smith, B. E., & Berthier, E. (2019). Ice shelf basal melt rates from a high resolution DEM record for Pine Island Glacier, Antarctica. *The Cryosphere*, 13, 2633–2656.
- Spence, P., Griffies, S. M., England, M. H., Hogg, A., Saenko, O. A., & Jourdain, N. C. (2014). Rapid subsurface warming and circulation changes of Antarctic coastal waters by poleward shifting winds. *Geophysical Research Letters*, 41, 4601–4610. https://doi.org/10.1002/ 2014GL060613
- Swart, N. C., Gille, S. T., Fyfe, J. C., & Gillett, N. P. (2018). Recent Southern Ocean warming and freshening driven by greenhouse gas emissions and ozone depletion. *Nature Geoscience*, 11, 838–841.
- Thomas, R. H. (2004). Force-perturbation analysis of recent thinning and acceleration of Jakobshavn Isbrae, Greenland. Journal of Glaciology, 50(168), 57–66.

Treasure, A. M., Roquet, F., Ansorge, I. J., Bester, M. N., Boehme, L., Bornemann, H., & de Bruyn P. J. N. (2017). Marine mammals exploring the oceans pole to pole: A review of the MEOP consortium. *Oceanography*, *30*, 132–138.

van Wessem, J. M., Reijmer, C. H., Morlighem, M., Mouginot, J., Rignot, M. B., Joughin, I., & Meijgaard, E. (2014). Improved representation of East Antarctic surface mass balance in a regional atmospheric climate model. *Journal of Glaciology*, 60, 761–770.

Wilson, N., Straneo, F., & Heimbach, P. (2017). Satellite-derived submarine melt rates and mass balance (2011–2015) for Greenland's largest remaining ice tongues. *The Cryosphere*, 11, 2773–2782.

Young, N. W., Goodwin, I. D., Hazelton, N. W. J., & Thwaites, R. J. (1989). Measured velocities and ice flow in Wilkes Land, Antarctica. Annals of Glaciology, 12, 192–197.

Young, D. A., Wright, A. P., Roberts, J. L., Warner, R. C., Young, N. W., & Greenbaum, J. S. (2011). A dynamic early East Antarctic Ice Sheet suggested by ice-covered fjord landscapes. *Nature*, 72–75.

Zwally, H. J., Schutz, R., Hancock, D., & Dimarzio, J. (2014). GLAS/ICESat L2 Global Land Surface Altimetry Data (HDF5), Version 34. https://doi.org/10.5067/ICESAT/GLAS/DATA211